Author's response to reviews

Title: Experiences of pharmacists involved in the delivery of a specialist asthma service in Australia

Authors:

Lynne M Emmerton (lynne.emmerton@curtin.edu.au)
Lorraine D Smith (lorraine.smith@sydney.edu.au)
Kate S LeMay (kate.lemay@sydney.edu.au)
Ines Krass (ines.krass@sydney.edu.au)
Bandana Saini (bandana.saini@sydney.edu.au)
Sinthia Z Bosnic-Anticevich (sinthia.bosnic-anticevich@sydney.edu.au)
Helen K Reddel (hkr@med.usyd.edu.au)
Deborah L Burton (d.burton@cqu.edu.au)
Kay Stewart (kay.stewart@monash.edu)
Carol L Armour (carol.armour@sydney.edu.au)

Version: 2 Date: 17 December 2011

Author's response to reviews:

1. The compliance of this study with the Declaration of Helsinki, and the names of the four Human Research Ethics Committees has been added to the Methods.

2. The Authors' Contributions have been edited.